Author’s response to reviews

Title: A Case of Pseudo-Wellens Syndrome, Acute Pancreatitis and an Anomalous Coronary Artery

Authors:
Valery Effoe (veffoe@msm.edu)
Wesley O'Neal (wesley.oneal@emory.edu)
Renato Santos (renato.santos@emory.edu)
Leon Rubinsztain (leon.rubinsztain@va.gov)
Abarmard Zafari (azafari@emory.edu)

Version: 1 Date: 12 Nov 2019

Author’s response to reviews:

We are grateful to both reviews and the excellent suggestions recommended by Reviewer 1.

Following are our responses to the comments by Reviewer 1:

1. The title of the case report should be reconsidered. Starting could be "A case of ---"
   We have changed the title accordingly, on page 1 (line 2) in red font to "A Case of Pseudo-Wellens Syndrome, Acute Pancreatitis and an Anomalous Coronary Artery".

2. It will be appreciated by readers if a paragraph regarding Wellens syndrome is included detailing this entity further.
   We have added a paragraph describing the Wellens syndrome in detail as suggested, on page 5 (lines 14-20) in red font.

3. While it is obvious that there there was no acute coronary syndrome, it will be important to include troponin trends since admission in hospital to clearly demonstrate lack of rising serum troponin over period of 8-24 hours. Currently the Table only includes Troponin level at ED. We appreciate the comment. Troponin I was checked twice on the same blood draw, once in the Emergency Department by point of care assay, and second by a high sensitivity troponin assay in the hospital’s central laboratory. A follow-up troponin assay was not rechecked once coronary angiography revealed angiographically normal coronary arteries and acute pancreatitis was diagnosed. A statement has been added to clarify the report on page 6 (line 20-23) in red font.